Author’s response to reviews

Title: Case Report of One month and 15 days old baby with Type V Hyperlipoproteinemia

Authors:

Dr. Shabnam dildar (shabnam.dildar@yahoo.com)

Tahir Shamsi (t.shamsi.62@gmail.com)

Version: 3 Date: 31 Jan 2020

Author’s response to reviews:

January 31st, 2020

Editor-in-Chief

Dr. Aldo Ferreira Hermosillo,

Subject: BEND-D-19-00409R2

Title: Case Report of One month and 15 days old baby boy with Type V Hyperlipoproteinemia

Dear Dr. Aldo Ferreira,

Thank you so much Sir, for your valuable comments for this case report. All comments and suggestion have been incorporated. All changes made are mentioned below

EDITOR COMMENTS:

1) Comment: add the specific triglyceride level and correct the grammar of the following sentence in the discussion section: “Our patient had a very severe hypertriglyceridemia with triglyceride levels > 885 mg/dl at the age of 1 month and 15 days, Cholestyramine therapy was started. His lipid profile repeated at the age of 3 months showed some improvements, his TGs reduced to 1986 mg/dl and TC to 105 mg/dl.”

1) Response: Specific triglyceride level added and grammar has corrected (in discussion section, line no 211-215)
2) Comment: Please also correct the grammar of the following sentence in the discussion section: “As per Gotoda T et al in 2012, in patients with type V hyperlipoproteinemia, the presence of underlying diseases or contributing factors such as diabetes, drinking was present in about 2/3 but not in the remaining 1/3, in this case environmental factors not involved for dyslipidemias development”. Please add the reference after the sentence: “The Endocrine Society does not recommend routine measurements of these tests for diagnosis of type HLP.”

2) Response: Grammar corrections done. Reference added (in discussion section, line no 230-233 and line 245-246)

3) Comment: Remove the “dedication section”. This section is not part of BMC Endocrine Disorders journal sections.

3) Response: done

4) Comment: corrects the references according submission guidelines.

4) Response: done